Author's response to reviews

Title: Epigenetic silencing of genes and microRNAs within the imprinted Dlk1-Dio3 region at human chromosome 14.32 in giant cell tumor of bone

Authors:

Burkhard Lehner (Burkhard.Lehner@med.uni-heidelberg.de)
Pierre Kunz (Pierre.Kunz@med.uni-heidelberg.de)
Heiner Saehr (Heiner.Saehr@med.uni-heidelberg.de)
Joerg Fellenberg (Joerg.Fellenberg@med.uni-heidelberg.de)

Version: 4 Date: 1 July 2014

Author's response to reviews:

Dear Editor,

we are pleased that our manuscript (MS:5210485601243857) is suitable for publication in BMC Cancer. At the beginning of the Methods section we already declared that informed consent was obtained from all patients. In the revised manuscript we changed this sentence to: “…..informed consent to analyze tumor tissue and to publish clinical details was obtained from all individuals….”. This part is highlighted in yellow in the revised manuscript.

Best regards

Jörg Fellenberg